Author's response to reviews

Title: The caCORE Software Development Kit: Streamlining construction of interoperable biomedical information services

Authors:

Joshua Phillips (joshua.a.phillips@comcast.net)
Ram Chilukuri (chilukur@mail.nih.gov)
Gilberto Fragoso (fragosog@mail.nih.gov)
Denise Warzel (warzeld@mail.nih.gov)
Peter A Covitz (covitzp@mail.nih.gov)

Version: 3 Date: 12 September 2005

Author's response to reviews:

Dear Editors,

This version of the manuscript has been edited as follows: Table 2 was edited by removing some less important columns to permit it to be properly displayed in "Portrait" page layout, and a footnote to the table was inserted to explain one of the column headers. The body text that describes Table 2 was edited slightly to better describe the revised table.

Sincerely,

Peter Covitz